Author’s response to reviews

Title: Exposure rate of VZV among women attending antenatal care clinic in Sri Lanka - a cross sectional study

Authors:

SWPL Daulagala (pavithra.daulagala@gmail.com)
Faseeha Noordeen (faseeha.noordeen12@gmail.com)
MMF Fara (fathina.mukthar@gmail.com)
Chathura Rathnayake (chathura67@hotmail.com)
Kapila Gunawardana (samodhar@yahoo.com)

Version: 2 Date: 08 Sep 2017

Author’s response to reviews:

Comment 1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors MMFF, CR, and KG are not clear, and do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead. Please note that any changes in the list of authors requires the completion of the “Change in authorship” form.

Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group - alone - does not usually justify authorship.

Response: We have justified authorship for those indicated by the journal.

Comment 2. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.
Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Response: Changes have been made as instructed by the journal.

Comment 3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Changes have been made as instructed by the journal.